Author’s response to reviews

Title: Developing a framework to evaluation knowledge into action interventions

Authors:
Sarah Morton (s.morton@ed.ac.uk)
Suzanne Wilson (suzannewilson@nhs.net)
Sheila Inglis (sheila@smciassociates.com)
Karen Ritchie (karenritchie@nhs.net)
Ann Wales (Ann.Wales@nes.scot.nhs.net)

Version: 3 Date: 08 Jan 2018

Author’s response to reviews:

1. Please reformat your in-text citations according to our submission guidelines:
   - The in-text citations should use the reference number in brackets, [1], rather than as superscript.
     Done

2. As indicated by our submission guidelines, please include a Conclusions heading in the text of your manuscript, as both a Conclusion and a Discussion section should be provided. You may consider placing it before the last paragraph of the Discussion section.
   Done

3. Please remove any figure titles and captions from the figure files as these files should contain the image graphics only. Upon doing this, please place the titles/captions at the end of the main manuscript after the References section under a newly created “Figure legend” heading.
   Done

4. Thank you for providing an Ethics and consent for participation heading. If consent for participation was not required, please provide details of why this was wavered and the institution/committee that approved this. If consent had been collected from participants then please clearly state so under the Ethics and consent for participation heading. When doing so,
please clarify whether informed consent obtained from participants was written or verbal. 
Detail added

5. Thank you for including a Funding heading for the Declarations section. Please list the full, unabbreviated name of the funding bodies that supported this research. Please also state the roles of the funding bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript. Detail added

6. Please remove the response to reviewers from the file inventory, as it is no longer needed at this stage of the editorial process. Removed

7. When submitting your revised manuscript please ensure you do so as a single clean copy without any tracked changes, colored or highlighted text, as these are no longer required at this stage of the editorial process. Done